Author's response to reviews

Title: Abscess of urachal remnants presenting with acute abdomen: a case series.

Authors:

Fadl M TAZI (mustaphaahsaini@hotmail.com)
Mustapha AHSAINI (drahsaini@gmail.com)
Abdelhak Khallouk (drahsaini@hotmail.com)
Soufiane MELLAS (sou3ad_123456@hotmail.com)
Roos E Stuurman-Wieringa (roos_w@hotmail.com)
Jamal M ELFASSI (ahsainimed@hotmail.fr)
My Hassan FARIH (brahim_a_01@hotmail.com)

Version: 4 Date: 18 April 2012

Author's response to reviews:

- All the modification of the style of the languages was done by a native english speaker.
- the modification of the figures with white coloured arrows with stems also was done.